\setlength\tabcolsep{3pt}
\begin{table*}[!htb]
\caption{Comparison of prior \aia{s}: attack vector exploited (e.g., $\targetmodel(X(\omega))$, $X(\omega)$, $Y(\omega)$, distribution over $S$ ($P_S$) and confusion matrix $C(Y,\targetmodel\circ X)$), whether $S$ is censored, i.e., included in $\traindata$ and inputs, whether \aia{s} account for class imbalance in $S$, whether \adv is active or passive and whether the threat model is blackbox or whitebox.}
\begin{center}
\footnotesize
% \resizebox{\textwidth}{!}{%
\begin{tabular}{ |c|c|c|c|c|c| }
 \hline
 \rowcolor{LightCyan} 
 \textbf{Literature} & \textbf{Attack Vector} & \textbf{Is $S$ censored?} & \textbf{Imbalance in $S$?} & \textbf{\adv} & \textbf{Threat Model} \\
 \hline
 \rowcolor{LightCyan} 
 \multicolumn{6}{|c|}{\textbf{Imputation-based Attacks}}\\
 \hline
 \textbf{Fredrikson et al.}~\cite{fredrikson2} & $X$, $Y$, $\targetmodel\circ X$, \textbf{$P_S$}, $C(Y,\targetmodel\circ X$) & $\checkmark$ & $\times$ &  Passive & Blackbox\\
 \textbf{Yeom et al.}~\cite{yeom} & $X$, $Y$, $\targetmodel$, \textbf{$P_S$} & $\checkmark$  & $\times$ &  Passive & Blackbox\\
  \textbf{Mehnaz et al.}~\cite{MehnazAttInf} & $X$, $Y$, $\targetmodel$, \textbf{$P_S$}, $C(Y,\targetmodel\circ X)$ & $\checkmark$ & $\times$ &  Passive & Blackbox\\
  \textbf{Jayaraman and Evans}~\cite{jayaraman2022attribute} & $X$, $Y$, $\targetmodel$, $P_S$, $C(Y,\targetmodel\circ X)$ & $\times$, $\checkmark$ & $\times$ &  Passive & Whitebox\\
  \hline
  \rowcolor{LightCyan} 
   \multicolumn{6}{|c|}{\textbf{Representation-based Attacks}}\\
  \hline
 \textbf{Song et al.}~\cite{Song2020Overlearning} & $\targetmodel\circ X$ & $\times$ & $\times$ &  Passive & Both\\
 \textbf{Mahajan et al.}~\cite{Mahajan2020DoesLS} & $\targetmodel\circ X$ & $\checkmark$ & $\times$ &  Passive & Blackbox\\
 \textbf{Malekzadeh et al.}~\cite{malekzadeh2021honestbutcurious} & $\targetmodel\circ X$ & $\times$ & $\times$ &  Active & Blackbox\\
 % \textbf{Our Work} & $\targetmodel\circ X$ & $\times$, $\checkmark$ & $\checkmark$ &  Passive & Blackbox \\
 \hline
\end{tabular}
% }
\end{center}
\label{tab:summary}
\end{table*}